Author's response to reviews

Title: Human papillomavirus (HPV) types 16, 18, 31, 45 DNA loads and HPV-16 integration in persistent and transient infections in young women.

Authors:

Agnihotram v. Ramanakumar (venkata.agnihotram@mcgill.ca)
Goncalves Otelinda (goncalves.otelinda.chum@ssss.gouv.qc.ca)
Harriet Richardson (HRichardson@ctg.queensu.ca)
Pierre Tellier (pierre-paul.tellier@mcgill.ca)
Alex Ferenczy (alex.ferenczy@mcgill.ca)
François Coutlée (francois.coutlee.chum@ssss.gouv.qc.ca)
Eduardo L Franco (eduardo.franco@mcgill.ca)

Version: 4 Date: 3 November 2010

Author's response to reviews:

Dear Editorial team,

We would like to resubmit our manuscript here based on suggested revisions in your email date 27th October 2010.

Major changes:

Author list: Authors names were corrected

Tables: The tables were incorporated in the text

Minor changes:

Email address: authors email were included in the title page,

Competing interests: statement on competing interests was deleted from he title page

Background: the sub-heading was added

Materials and methods: renamed as 'Methods'

Highlighting and tracking: removed from the text.

Figures: adjusted our best. We leave it to Editorial official to shape them, and crop them more.

Typography: checked for topographical errors.

Please let us know if we have to make any changes.

Authors